Author's response to reviews

Title: HIF-1 activation induces doxorubicin resistance in MCF7 3-D spheroids via P-glycoprotein expression: a potential model of the chemo-resistance of invasive micropapillary carcinoma of the breast

Authors:

Sophie Doublier (sophie.doublier@unito.it)
Dimas C Belisario (caro.belisario@gmail.com)
Manuela Polimeni (manuela.polimeni@unito.it)
Laura Annaratone (laura.annaratone@unito.it)
Chiara Riganti (chiara.riganti@unito.it)
Elena Allia (elena.allia@unito.it)
Dario Ghigo (dario.ghigo@unito.it)
Amalia Bosia (amalia.bosia@unito.it)
Anna Sapino (anna.sapino@unito.it)

Version: 4 Date: 25 November 2011

Author's response to reviews: see over
Torino, November 25, 2011

To BMC Cancer Editorial board,

we thank the Editor and the Reviewers for the comments and the final decision concerning our manuscript entitled “HIF-1α activation induces doxorubicin resistance in MCF7 3-D spheroids via P-glycoprotein expression: a potential model of the chemo-resistance of invasive micropapillary carcinoma of the breast” (MS: 2072559323540958). As requested by the Editor, we have now made the correction of the grammatical/typographical/nomenclature concerns described by Reviewer 3. A copy of the manuscript containing the changes highlighted in red has been included in our re-submission (pages 6, 10, 14, 33 and 34).

In the version 2 of the manuscript, we have modified the title: “HIF-1 activation induces doxorubicin resistance in MCF7 3-D spheroids via P-glycoprotein expression: a potential model of the chemo-resistance of invasive micropapillary carcinoma of the breast” and the abstract in the Manuscript details tab. However, these changes were not taken in consideration. We updated them again in the Manuscript details tab even if between version 2 and version 3 of the manuscript, title and abstract were not modified.

Thanking in advance for your time and concern,

Sophie Doublier
Corresponding author: Dr. Sophie Doublier, Dipartimento di Genetica, Biologia e Biochimica, Via Santena 5/bis, 10126 Torino, Italy - Phone: +39-0116705843; fax: +39-0116705845; e-mail: sophie.doublier@unito.it